\documentclass[12pt,reqno]{amsart}
\usepackage{
  amsmath, amssymb, multicol, mathtools, stmaryrd, xcolor,
  mathabx, needspace, lgreek, sidecap, %mathbbol,
  blindtext, dbnsymb
}
\usepackage[export]{adjustbox} % This also includes graphicx.
\usepackage{enumitem}
\usepackage[all]{xy}
%\usepackage[english,greek]{babel}
\usepackage[textwidth=6.5in,textheight=9in,headsep=0.15in,centering]{geometry}

% Following http://tex.stackexchange.com/a/847/22475:
\usepackage[breaklinks=true]{hyperref}\hypersetup{colorlinks,
  linkcolor={green!50!black},
  citecolor={green!50!black},
  urlcolor=blue
}

\usepackage{caption}
\captionsetup[figure]{textfont=sf,width=0.9\linewidth}

\usepackage[stable]{footmisc} % to allow a footnote within a section title.

\begin{document}
\begin{thebibliography}{BDHLS}

\bibitem[AKKN]{AKKN} A.~Alekseev, N.~Kawazumi, Y.~Kuno, and F.~Naef,
  {\em The goldman–turaev lie bialgebra in genus zero and the kashiwara–vergne problem,}
  Advances in Mathematics {\bf 326} (2018) 1-–53, \arXiv{1703.05813}.

\bibitem[AM]{AlekseevMeinrenken:KV} A.~Alekseev and E.~Meinrenken,
  {\em On the Kashiwara-Vergne conjecture,}
   Inventiones Mathematicae, {\bf 164} (2006) 615--634, \arXiv{math/0506499}.

\bibitem[AT]{AlekseevTorossian:KashiwaraVergne} A.~Alekseev and C.~Torossian,
  {\em The Kashiwara-Vergne conjecture and Drinfel'd's associators,}
   Annals of Mathematics, {\bf 175} (2012) 415--463, \arXiv{0802.4300}.

\bibitem[BN1]{Bar-Natan:OnVassiliev} D.~Bar-Natan,
  \href{https://www.math.toronto.edu/~drorbn/LOP.html#OnVassiliev}{{\em
    On the Vassiliev knot invariants,}}
  Topology {\bf 34} (1995) 423--472.

\bibitem[BN2]{Bracelets} D.~Bar-Natan,
  \href{https://www.math.toronto.edu/~drorbn/papers/Bracelets}{{\em Bracelets and the Goussarov Filtration of the Space of Knots,}}
  Invariants of knots and 3-manifolds (Kyoto 2001), Geometry and Topology Monographs {\bf 4} 1--12, \arXiv{https://arxiv.org/abs/math/0111267}.

\bibitem[BN3]{LesDiablerets-2208} D.~Bar-Natan,
  {\em Tangles in a Pole Dance Studio: A Reading of Massuyeau, Alekseev, and Naef,}
  talk given in {\em Algebra, Topology and the Grothendieck-Teichmüller group}, Les Diablerets, September 2022. Handout and video at \url{http://drorbn.net/ld22}.

\bibitem[BD]{WKO2} D.~Bar-Natan and Z.~Dancso,
  {\em Finite Type Invariants of W-Knotted objects II: Tangles, Foams and the Kashiwara-Vergne Problem}
  Math.\ Annalen {\bf 367} (2017) 1517--1586, \arXiv{1405.1955}.

\bibitem[BDHLS]{BDHLS} D.~Bar-Natan, Z.~Dancso, T.~Hogan, J.~Liu, and N.~Scherich,
  {\em Goldman-Turaev formality from the Kontsevitch integral,}
  in preparation.

\bibitem[CDM]{ChmutovDuzhinMostovoy:Vassiliev} S.~Chmutov, S.~Duzhin, and J.~Mostovoy,
  {\em Introduction to Vassiliev Knot Invariants,}
  Cambridge University Press, Cambridge UK, 2012.

\bibitem[KR]{KhovanovRozansky:MatrixFactorization} M.~Khovanov and L.~Rozansky,
  {\em Matrix Factorizations and Link Homology,}
  Fundamenta Mathematicae {\bf 199} (2008) 1--91, \arXiv{math/0401268}.

\bibitem[KV]{KashiwaraVergne:Conjecture} M.~Kashiwara and M.~Vergne,
  \href{http://www.springerlink.com/content/v73014gx14084624/}{{\em The
    Campbell-Hausdorff Formula and Invariant Hyperfunctions,}}
  Invent.\ Math.\ {\bf 47} (1978) 249--272.

\bibitem[Ku]{Kuno:EmergentVersion} Y.~Kuno,
  {\em Emergent Version of Drinfeld's Associator Equations,}
  \arXiv{2504.02549}.

\bibitem[Po]{Polyak:Reidemeister} M.~Polyak,
  {\em Minimal generating sets of Reidemeister moves,}
  Quantum Topol.\ {\bf 1} (2010) 399-–411, \arXiv{0908.3127}.

\end{thebibliography}
 %\latintext
\newdimen\captionwidth\captionwidth=\hsize

\title{
  Questions Asked at a Banff Conference, April 2025.
}

\author{Dror~Bar-Natan}

\date{First edition June 12, 2025. This edition \today. See also \url{http://drorbn.net/ba25}.}

\maketitle

\section{Question 1: Emergent Knotted Objects}

\def\bbQ{{\mathbb Q}}
\def\bbR{{\mathbb R}}
\def\calA{{\mathcal A}}
\def\calD{{\mathcal D}}
\def\calI{{\mathcal I}}
\def\calK{{\mathcal K}}
\def\calU{{\mathcal U}}
\def\dk{{\mathsf{dk}}}
\def\edk{{\mathsf{edk}}}

\def\PB{{\mathit{PB}}}
\def\PDS{{\mathit{PDS}}}

\def\gr{\operatorname{gr}}

\def\arXiv#1{{\href{http://arxiv.org/abs/#1}{{arXiv:}\linebreak[0]{#1}}}}

Consider some space $\calK$ of knotted objects in a three manifold $M$ (knots, or links, or knotted graphs, or tangles if $M$ has a boundary, or braids if one can specify a ``vertical'' direction in $M$). If we mod out $\calK$ by the relation $\overcrossing=\undercrossing$, knottedness is eliminated and what remains is a space of curves (or multi-curves, or graphs, etc.) regarded up to homotopy. ``Emergent knotted objects'' are what you get when you mod out $\bbQ\calK$, the space of formal linear combinations of elements of $\calK$, by a slightly weaker version of $\overcrossing=\undercrossing$ known in the theory of finite type invariants~\cite{Bar-Natan:OnVassiliev, ChmutovDuzhinMostovoy:Vassiliev} as the relation $\doublepoint^2=0$, or more precisely, $(\doublepoint\doublepoint) \coloneqq (\overcrossing\overcrossing) - (\overcrossing\undercrossing) - (\undercrossing\overcrossing) + (\undercrossing\undercrossing) = 0$. Knottedness almost entirely disappears yet in the spaces we get, $\calK^{em}\coloneqq\bbQ\calK/(\doublepoint^2=0)$, we see that just a bit of knot theory begins to emerge beyond the homotopy theory that is already there. If, for example, we are looking at links in $\bbR^3$, then the only non-constant invariants in $(\calK^{em})^*$ are the linking numbers.

{\bf Questions.} What can you say about emergent knots in general? How much more are they than just homotopy classes?

Emergent knots can be filtered using the Goussarov bracelet filtration~\cite{Bracelets} (it makes sense in an arbitrary three manifold, and it is not the same as the Vassiliev filtration as the different ``detours'' may differ in homotopy).

{\bf Questions.} What is the associated graded $\gr\calK^{em}$ of $\calK^{em}$? Are there expansions (universal finite type invariants) $Z^{em}\colon\calK^{em}\to\gr\calK^{em}$?

In the case where $M$ is a pole dancing studio ($D^2\times I$ with a number of vertical lines removed), the references \cite{LesDiablerets-2208, BDHLS, Kuno:EmergentVersion} suggest strongly that the answers to these questions are closely related to the much-studied and highly non-trivial ``Kashiwara-Vergne equations'' \cite{KashiwaraVergne:Conjecture, AlekseevMeinrenken:KV, AlekseevTorossian:KashiwaraVergne, WKO2, AKKN}.

The appendix of \cite{Kuno:EmergentVersion} is a more detailed version of this question.

\section{Question 2: Knots and Bnots}

After asking this question I learned that it is essentially contained in Polyak's lovely paper~\cite{Polyak:Reidemeister} (see especially his Figure~7 and the discussion around it), and so I will keep my discussion brief. We all know that the set of knots is equivalent to the set $\calD$ of knot diagrams, modulo the three Reidemeister moves R1, R2, and R3, taken with all possible strand orientations. The moves R2 and R3 have two variants: The ``b'' variant which stands for ``those R2 and R3 moves that may occur in braids'', and the ``c'' variant, which stands for ``those moves in which the middle bigon/triangle is cyclically oriented''. So more precisely, $\{\text{knots}\}=\calD/(R1,R2b,R2c,R3b,R3c)$. We define $\{\text{bnots}\}\coloneqq\calD/(R1,R2b,R3b)$. Namely, bnots are knot diagrams module R1 moves and only the R2 and R3 moves that may occur in braids.

The ``obvious'' (False) map $F\colon\{\text{knots}\}\to\{\text{bnots}\}$ which maps every knot diagram to itself regarded as a bnot diagram is not well defined because its domain has more relations than its range. And yet there is a more subtle (True!) embedding $T\colon\{\text{knots}\}\to\{\text{bnots}\}$: namely, for any knot $K$ find a braid $B$ whose closure is $K$ and map $T\colon K\mapsto B$. This map is well defined because the moves that occur within Markov's theorem, which relates knots and braids, are only R1 and the ``b'' versions of R2 and R3. I expect that there are many bnots that are not in the image of the map $T$.

{\bf Question.} What are bnots? How much more are they than knots? Do they have any significance?

In principle, there may be quantities that can be defined on all knot diagrams and are invariant only under R1, R2b, and R3b. Such quantities, even though they are defined on all knot diagrams, define knot invariants only through the use of braids. I know of only one real life example: Khovanov-Rozansky homology~\cite{KhovanovRozansky:MatrixFactorization}.



\end{document}
\endinput
